Author's response to reviews

Title: Effectiveness of diclofenac versus acetaminophen in primary care patients with knee osteoarthritis: [NTR1485], DIPA-Trial: design of a randomized clinical trial

Authors:

Saskia PJ Verkleij (s.verkleij@erasusmc.nl)
Pim AJ Luijsterburg (p.luijsterburg@erasusmc.nl)
Bart W Koes (b.koes@erasusmc.nl)
Arthur M Bohnen (a.bohnen@erasusmc.nl)
Sita MA Bierma-Zeinstra (s.bierma-zeinstra@erasusmc.nl)

Version: 2 Date: 5 January 2010

Author's response to reviews: see over
Rotterdam, 5th January 2010

Dear Editor,

Thank you for your decision to publish our manuscript.

Please find enclosed the resubmission of our design article entitled ‘Effectiveness of diclofenac versus acetaminophen in primary care patients with knee osteoarthritis: [NTR1485], DIPA-Trial: design of a randomized clinical trial’.

Referring to your email of December the 18th, we have made the following changes to our manuscript:

We have corrected the structure of our abstract and manuscript.
We have deleted the number of references and table/figures from page 1.
We have cropped our figure as closely as possible and removed the title and figure number. Besides that, we have enclosed the legend and title of the figure to our manuscript on page 18.
Finally, we checked our manuscript for errors and made some final corrections and revisions.

We hope that the changes we have made are sufficient and we look forward to the publication of our manuscript in *BMC Musculoskeletal Disorders*.

Yours sincerely,

On behalf of all authors,

Saskia Verkleij

Address correspondence to: Saskia PJ Verkleij, Dept. of General Practice, Erasmus MC, University Medical Centre Rotterdam, the Netherlands. Tel: +31 10 703 2 117. Fax: +31 10 703 2 127. E-mail: s.verkleij@erasusmc.nl